Author’s response to reviews

Title: ASSESSMENT OF POSTOPERATIVE PAIN MANAGEMENT AND COMPARISON OF EFFECTIVENESS OF PAIN RELIEF TREATMENT INVOLVING PARAVERTEBRAL BLOCK AND THORACIC EPIDURAL ANALGESIA IN PATIENTS UNDERGOING POSTEROLATERAL THORACOTOMY

Authors:

Małgorzata Wojtyś (margaretkaw@wp.pl)
Józef Wąsikowski (jw2002@o2.pl)
Norbert Wójcik (noert@wp.pl)
Janusz Wójcik (janusz.zenon.wojcik@wp.pl)
Piotr Wasilewski (wapiotr@interia.pl)
Piotr Lisowski (piotr97@gmail.com)
Tomasz Grodzki (grodzki@grodzki.szczecin.pl)

Version: 2 Date: 03 Feb 2019

Author’s response to reviews:

Szczecin, 03 February 2019

Journal of Cardiothoracic Surgery

Response to the editor's review of an article entitled "ASSESSMENT OF POSTOPERATIVE PAIN MANAGEMENT AND COMPARISON OF EFFECTIVENESS OF PAIN RELIEF TREATMENT INVOLVING PARAVERTEBRAL BLOCK AND THORACIC EPIDURAL ANALGESIA IN PATIENTS UNDERGOING POSTEROLATERAL THORACOTOMY." written by M.E. Wojtyś et al. from the Department of Thoracic Surgery and Transplantation of the Pomeranian Medical University, Szczecin, and the Department of Anesthesiology and Intensive Therapy of Independent Public Provincial Hospitals in Szczecin, which was submitted with the request for publication.

I would like to begin by expressing the authors' acknowledgements for the devoted time and valuable remarks. Authors of the said article fully agree with the Reviewers comments.
Having familiarized themselves with the Reviewers remarks, the authors used their best efforts to comply with reviewer’s requests.

The study involved 2 groups of patients, each consisting of 30 patients. The amount of patients was limited due to the duration of the study, although the results of this study will be useful in establishment of postoperative pain management in patients undergoing posterolateral thoracotomy.

Being aware of the work's restrictions, authors of the article used their best efforts to fulfill the requirements applicable for scientific articles published in Your journal. Having taken into account the Editor's remarks, acting in my own name and in the name of the remaining authors, I would like to kindly request publication of the said work in Your journal.

Best regards

Małgorzata Wojtyś